Author's response to reviews

Title: Clinical presentation and predictors of outcome in patients with severe acute exacerbation of chronic obstructive pulmonary disease requiring admission to intensive care unit

Authors:

Alladi Mohan (alladimohan@rediffmail.com)
Raya Premanand (prema2@sancharnet.in)
Lebaka Narayana Reddy (lebaka2003@yahoo.co.in)
Mangu H Rao (drmhrao@yahoo.com)
Surendra K Sharma (sksharma@aiims.ac.in)
Ranjit Kamity (ranjith.kamity@gmail.com)
Srinivas Bollineni (shsri_b@yahoo.co.in)

Version: 7  Date: 18 December 2006

Author's response to reviews:

All the formatting changes mentioned in e-mail dated 12 Dec 2006, have been incorporated.

Author list has been written in the prescribed format (forename,initial (if any), surname) identically and in the order in which they are listed in the submission system.

Title & Abstract have been checked to ensure that these are identical in the manuscript and in the submission system.

Section headings have been formatted in sentence case as instructed.

Keywords section has been removed.

"Introduction" has been renamed as "Background" as instructed.

Footnotes have been deleted and as per the suggestion, the presentation details have been listed under "Acknowledgements".

Tables 4a and 4b have been renumbered as Tables 4 and 5 and any references to the tables in the text have been updated to reflect these changes.

Inadvertant typing errors have been corrected.